



September 30, 2013

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 5374-edited.-Licata et al.doc).

Title: CLINICAL COURSE AND PROGNOSTIC FACTORS OF HEPATORENAL SYNDROME: A RETROSPECTIVE SINGLE CENTER COHORT STUDY

Author: Analisa Licata, Marcello Maida, Ambra Bonaccorso, Fabio Salvatore Macaluso, Maria Cappello, Antonio Craxì, Piero Luigi Almasio.

Name of Journal: *World Journal of Hepatology*

ESPS Manuscript NO: 5374

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

Thank you again for publishing our manuscript in the *World Journal of Hepatology*.

Sincerely yours,

Piero L. Almasio, Section of Gastroenterology, Di.Bi.M.I.S., University of Palermo, Piazza delle Cliniche, 2, 90127 Palermo, Italy.

E-mail: piero.almasio@unipa.it